\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{???.??} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{??X??} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}




\begin{document}
\begin{thebibliography}{1} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).
\bibitem{1} EM. Matveev, \href{
	https://doi.org/10.1070/im2000v064n06abeh000314}{\textit{An explicit lower bound for a homogeneous rational linear form in the logarithms of algebraic numbers}}, II, Izv. Ross. Akad. Nauk Ser. Mat. \textbf{64} (2000), 125-180. Translation in Izv. Math. \textbf{64} (2000), 1217-1269.

\bibitem{2} Dujella A, Petho A, \href{https://doi.org/10.1093/qmathj/49.3.291}{\textit{A generalization of a theorem of Baker and Davenport}}. The Quarterly Journal of Mathematics, \textbf{49} (3) (1998), 291-306.

\bibitem{3} Y. Bugeaud, M. Mignotte and S. Siksek, \href{https://doi.org/10.4007/annals.2006.163.969}{\textit{Classical and modular approaches to exponential Diophantine equations I. Fibonacci and Lucas perfect powers}}, Ann. of Math. \textbf{163} (3) (2006), 969-1018.
\bibitem{4} D. Marques and A. Togb\`e, \href{https://doi.org/10.3792/pjaa.89.47}{\textit{Fibonacci and Lucas numbers of the form $2^{a}+3^{b}+5^{c}$}}, Proc. Japan Acad. \textbf{89} (3) 2013, 47-50.
\bibitem{5} Y. Qu, J. Zeng and Y. Cao, \href{https://doi.org/10.3390/sym10100509}{\textit{Fibonacci and Lucas Numbers of the Form $2^{a}+3^{b}+5^{c}+7^{d}$}}, Symmetry. \textbf{10} (509) 2018, 
\bibitem{6} K. Chi Chim and V. Ziegler,  \textit{On Diophantine equation Involving Sum of Fibonacci numbers and Powers of $2$}, INTEGERS. \textbf{18} (A99) 2018.
\bibitem{7} N. Irmak and B. He, \href{doi: 10.7546/nntdm.2019.25.4.102-109.}{\textit{s-th power of Fibonacci number of the form $2^{a} + 3^{b} + 5^{c}$}}, Note on Number Theory and Discrete Mathematics. \textbf{25} (4) 2019, 102-109.
\bibitem{8} Y. Qu and J. Zeng, \href{10.21136/CMJ.2019.0265-18}{\textit{Pell and Pell-Lucas Numbers of the Form $-2^{a}-3^{b}+5^{c}$}}, Czechoslovak Mathematical Journal. \textbf{70} (1) 2020, 281-289.
\bibitem{9} K. Bhoi and P.Ray, \href{https://doi.org/10.46298/cm.10476}{ \textit{On the Diophantine equation $B_{n_{1}} + B_{n_{2}} = 2^{a_{1}} + 2^{a_{2}} + 2^{a_{3}}$}}, Communications in Mathematics. \textbf{31} (1) 2023, 375-392
\bibitem{10} A. Baker and H. Davenport, \href{https://doi.org/10.1093/qmath/20.1.129}{\textit{The equations $3x^2 - 2 = y^2$ and $8x^2 - 7 = z^2$}}, Quart. J. Math. Oxford Ser. \textbf{20} (1969), 129--137.
\bibitem{11} A. Baker and G. W\"{u}stholz, \href{https://doi.org/10.1515/crll.1993.442.19}{\textit{Logarithmic forms and group varieties}}, J. Reine Angew. Math. \textbf{442} (1993), 19--62.
\bibitem{12} J. J. Bravo and F. Luca, \href{https://doi.org/10.2989/16073606.2015.1070377}{\textit{On the Diophantine equation $F_{n}+F_{m}=2^{a}$}}, Quaest. Math. J. \textbf{39} (3) (2016), 391-400.
\bibitem{13} A. Behera and G. K. Panda, \textit{On the square roots of triangular numbers}. Fibonacci Quart. \textbf{37} (2) (1999), 98-105.


\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[$B_{r}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$, $q_{s}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$ ]{\large Balancing and Modified Pell numbers of the form $-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$}


\author[HUNAR SHERZAD TAHER]{{\bf HUNAR SHERZAD TAHER}\protect\orcidicon{0000-0002-1830-4032}}
\author[SAROJ KUMAR DASH]{{\bf SAROJ KUMAR DASH}\protect\orcidicon{0000-0001-8620-110X}}


\address{HUNAR SHERZAD TAHER AND SAROJ KUMAR DASH
\newline\hphantom{iii} Mathematics Division, School of Advance Science,
\newline\hphantom{iii} Vellore Institute of Technology, Chennai Campus,
\newline\hphantom{iii} 600127, Chennai, India.}
\email{\textcolor{blue}{hunarsherzad.taher2022@vitstudent.ac.in, sarojkumar.dash@vit.ac.in}}

\thanks{\sc Hunar Sherzad Taher, Saroj Kumar Dash,
Balancing and Modified Pell numbers of the form $-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$}
\thanks{\copyright \ 2023  Hunar Sherzad Taher, Saroj Kumar Dash}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc }
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} In this paper, we investigate the solutions of Balancing and Modified Pell numbers of the form $-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$, where $a_{1},a_{2},a_{3},a_{4},a_{5}$, are non-negative integers with $0\leq\max\{a_{1},a_{2},a_{3},a_{4}\}\leq a_{5}$.\medskip

\noindent{\bf Keywords:} Exponential Diophantine equation, Linear forms in logarithms, Modified Pell number, Balancing number.
 \end{quote}
}

\bigskip

\section{Introduction}
The balancing sequence $\left\{B_r\right\}_{r \geq 1}$ is derived from Panda and Behera's \cite{13} simple Diophantine equation 
$$
1+2+\ldots+(r-1)=(r+1)+(r+2)+\cdots+(r+t),
$$ A balancer corresponding to $r$ is represented by the integer $t$. For any $r\geq2$ the recurrence of the Balancing sequence is determined by $B_{r}=6B_{r-1}-B_{r-2}$, where $B_{0}=0$ and $B_{1}=1$. For any  $s\geq2$, the Modified Pell sequence is defined as $q_{s}=2 q_{s-1}+q_{s-2}$, where $q_{0}=q_{1}=1$. Many recent studies have focused on Diophantine equations linked to binary recurrence sequences and perfect powers, either for specified unique bases or the same bases for different powers. Fibonacci and Lucas numbers of the pattern $2^{a}+3^{b}+5^{c}$, where $a,b,c$, are non-negative integers with $c\geq\max\{a,b\}\geq0$, were discovered by Marques and Togb? Bravo \cite{4}. Bravo and Luca \cite{12} discussed how to solve the Diophantine equation $F_{n}+F_{m}=2^{a}$
where $n,m$ and $a$ are all positive integers. Qu, Zeng, and Cao \cite{5} proved Fibonacci and Lucas numbers of the form $2^{a}+3^{b}+5^{c}+7^{d}$  where $a,b,c,d$ are non-negative integers with $d\geq\max\{a,b,c\}$.\\
Furthermore, Chi Chim and Ziegler\cite{6} demonstrated the sum of Fibonacci numbers as a sum Powers of $2$. Irmak and He \cite{7} ] discovered the solution to the s-th power of the Fibonacci number of the form $2^{a} + 3^{b} + 5^{c}$, where $a,b,c$ and $s$ are positive integers with $1\leq\max\{a,b\}\leq c$. Recently, Qu and Zeng\cite{8} demonstrated that Pell and Pell-Lucas Numbers of the Form  $-2^{a}-3^{b}+5^{c}$, where $a,b,c$ are non-negative integers with $c\geq \max\{a,b\}\geq0$. In 2023, Bhoi and Ray \cite{9} find all solutions of the Diophantine equation $B_{n_{1}} + B_{n_{2}} = 2^{a_{1}} + 2^{a_{2}} + 2^{a_{3}}$, for positive integers $n_{1},n_{2},a_{1},a_{2},a_{3}$. In this study, we tend to demonstrate all solutions of the Balancing and Modified Pell numbers of the type $-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}$,
using the same technique as in the previous results.
\section{Auxiliary results}
\subsection{Balancing sequence}
We have the Balancing sequence’s Binet formula is
\begin{equation}\label{eq2.1}
	B(r)=\frac{\alpha^{r}-\beta^{r}}{4\sqrt{2}}\text { for all } r \geq 0,
\end{equation}
where $\beta=3-\sqrt{8}$ and $\alpha=3+\sqrt{8}$ are roots of the polynomial 
\begin{equation}
	f(x)=x^{2}-6 x+1.\nonumber
\end{equation}
The inequality holds true for the all-natural integer 
\begin{equation}\label{eq2.2}
	\alpha^{r-1} \leq B_{r} \leq \alpha^{r}.
\end{equation}
\subsection{Modified Pell sequence}
\label{Subsec:1}
The Binet formula of the Modified Pell sequence is 
\begin{equation}\label{eq2.3}
	q_{s}=\frac{\lambda^{s}+\delta^{s}}{\lambda+\delta} \text { for all } s \geq 0,
\end{equation}
where $\delta=1-\sqrt{2}$ and $\lambda=1+\sqrt{2}$ are roots of the polynomial
\begin{equation}
	f(x)=x^{2}-2 x-1.\nonumber
\end{equation}
The inequality can be easily shown by induction, and it holds for the all-natural number $s$. 
\begin{equation}\label{eq2.4}
	\lambda^{s-1} \leq q_{s} \leq \lambda^{s}.
\end{equation}
\subsection{Linear forms in logarithmic}
\begin{definition}(Absolute logarithmic height)\\
	Let  $\gamma$ be an algebraic number in the number field  $\mathbb{K}$ of degree $d$ with minimal polynomial on $\mathbb{Z}$
	\begin{equation}
		c_{0} x^{d}+c_{1} x^{d-1}+\ldots+c_{d}=c_{0} \prod_{i=1}^{d}\left(x-\gamma^{(i)}\right),\nonumber
	\end{equation}
	where each $\gamma^{(i)}$ is a conjugate of $\gamma$, and $c_{i}$ 's are relatively primes to each other with $c_{0}>0$. Then the logarithmic height of $\gamma$ is defined by:
	\begin{equation}
		h(\gamma)=\frac{1}{d}\left(\log c_{0}+\sum_{i=1}^{d} \log \left(\max \left\{\left|\gamma^{(i)}\right|, 1\right\}\right)\right) .
	\end{equation}
\end{definition} 
The logarithmic height's following characteristics can be demonstrated: 
\begin{equation}
	h(\eta \pm \gamma) \leq h(\eta)+h(\gamma)+\log 2,
\end{equation}
\begin{equation}
	h\left(\eta \gamma^{ \pm 1}\right) \leq h(\eta)+h(\gamma),
\end{equation}
\begin{equation}
	h\left(\eta^{k}\right)=|k| h(\eta).
\end{equation}
In order to demonstrate Theorems $\ref{th2.4}$ and $\ref{th2.5}$, we use Baker and  W\"{u}stholz \cite{11} lower bound for linear forms in algebraic number logarithms. Using a modified version of the Matveev result \cite{1}, we are the following \cite[Theorem 9.4]{3}.
\begin{theorem}\label{th2.2}
	Let $\mathbb{L}$ be a real algebraic number field of degree $D$ over $\mathbb{Q}$. Let $\gamma_{1}, \ldots, \gamma_{t}\in \mathbb{L}$ be a positive real algebraic numbers and $b_{1}, b_{2}, \ldots, b_{t}$ be nonzero integers, with
	\begin{equation}
		\Lambda:=\gamma_{1}^{b_{1}} \cdots \gamma_{t}^{b_{t}}-1,\nonumber
	\end{equation}
	is not zero. Then
	\begin{equation}
		\log |\Lambda|>G\left(A_{1} \ldots A_{t}\right)(1+\log D)(1+\log B),\nonumber
	\end{equation}
	where $G=(-1.4)\left(30^{t+3}\right)\left(t^{4.5}\right)\left(D^{2}\right)$, and
	\begin{equation}
		B \geq \max \left\{\left|b_{1}\right|, \ldots,\left|b_{t}\right|\right\},\nonumber
	\end{equation}
	and
	\begin{equation}
		A_{i} \geq \max \left\{D h\left(\gamma_{i}\right),\left|\log \left(\gamma_{i}\right)\right|, 0.16\right\}, 1 \leq i \leq t.\nonumber
	\end{equation}
	where $A_{1}, \ldots, A_{t}$ are real numbers.
\end{theorem} 
Determining the distance of a real number $T$ to the closest integer is expressed as $\|T\|=$ $\min \{|T-n|: n \in \mathbb{Z}\}$ .\\

The following lemma, which is an extension of the well-known Baker Davenport lemma \cite{10} and was proven by Dujella and Petho \cite{2}, is used to lower an upper bound.
\begin{lemma}\label{lem2.3}
	If $M$ is a positive integer and $p/q$ is a convergent continued fraction of the irrational number $\gamma$ such that $q>6 M$, let $K, L, \mu$ be real numbers with $K>0$ and $L>1$ and $\varepsilon:=\|\mu q\|-M\|\gamma q\|$. Then the inequality has no solution if $\varepsilon>0$,
	\begin{equation}
		0<\left|u \gamma-r+\mu\right|<\frac{K}{L^{w}},\nonumber
	\end{equation}
	in positive integers $u, r$ and $w$, with
	\begin{equation}
		u \leq M \text { and } w \geq \frac{\log (K q / \varepsilon)}{\log L}.\nonumber
	\end{equation}
\end{lemma}
\begin{theorem}\label{th2.4} The non-negative integer solutions $(r,a_{1},a_{2},a_{3},a_{4},a_{5})$ of the Diophantine equation \begin{equation}\label{eq2.9}
		B_{r}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}},
	\end{equation} with $a_{5}\geq\max\{a_{1},a_{2},a_{3},a_{4}\}$ are $(r,a_{1},a_{2},a_{3},a_{4},a_{5})\in \{(0, 1, 0, 0, 1, 1),\\(0, 1, 1, 1, 0, 1),(1, 0, 0, 0, 1, 1),(1, 0, 1, 1, 0, 1),(2, 1, 0, 0, 0, 1),(5, 0, 2, 3, 1, 3)\}$.
\end{theorem} 
\begin{theorem}\label{th2.5} The non-negative integer solutions $(r,a_{1},a_{2},a_{3},a_{4},a_{5})$ of the Diophantine equation \begin{equation}\label{eq2.10}
		q_{s}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}},
	\end{equation}with $a_{5}\geq\max\{a_{1},a_{2},a_{3},a_{4}\}$ are $(r,a_{1},a_{2},a_{3},a_{4},a_{5})\in \{(0, 0, 0, 0, 1, 1),\\(0, 0, 1, 1, 0, 1),(1, 0, 0, 0, 1, 1),(1, 0, 1, 1, 0, 1),(2, 0, 0, 1, 0, 1),(3, 0, 0, 0, 0, 1),\\(6, 0, 2, 1, 1, 2)\}$.
\end{theorem} 
\section{Proof theorem $\ref{th2.4}$}
\subsection{Bounding r}\label{sub 1}
We are combining $\eqref{eq2.1}$ and $\eqref{eq2.9}$, we get
\begin{equation}
	\frac{\alpha^r}{4\sqrt{2}}-11^{a_{5}}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+\frac{\beta^{r}}{4\sqrt{2}},\nonumber
\end{equation}
Taking absolute values for both sides, we get
\begin{equation}
	\left|\frac{\alpha^r}{4\sqrt{2}}-11^{a_{5}}\right|<2^{a_{1}}+3^{a_{2}}+5^{a_{3}}+7^{a_{4}}+\frac{1}{8\sqrt{2}},\nonumber
\end{equation}
because $|\beta|^{r}<\frac{1}{2}$. Dividing both sides by $11^{a_{5}}$ thus
\begin{equation}
	\left|\frac{\alpha^{r}11^{-a_{5}}}{4\sqrt{2}}-1\right|<\frac{2^{a_{1}}}{11^{a_{5}}}+\frac{3^{a_{2}}}{11^{a_{5}}}+\frac{5^{a_{3}}}{11^{a_{5}}}+\frac{7^{a_{4}}}{11^{a_{5}}}+\frac{1}{8\sqrt{2}11^{a_{5}}},\nonumber
\end{equation}
yields
\begin{equation}\label{eq3.1}
	\left|\frac{\alpha^{r}11^{-a_{5}}}{4\sqrt{2}}-1\right|<\frac{5}{11^{0.15a_{5}}}.
\end{equation}
As a result of the inequality $\eqref{eq2.2}$ and equation $\eqref{eq2.9}$, we now want to determine the relationship between  $r$ and $a_{5}$.
\begin{equation}
	\alpha^{r-1} \leq B_{r}<11^{a_{5}},\nonumber
\end{equation}
which yields $a_{5}>0.74r-0.74$ and $r<1.37a_{5}+1$. If $a_{5} \leq 15 $, then $r< 22 $.  In order to find the solutions to Theorem $\ref{th2.4}$, we use Maple programming to search in $0\leq a_{5}\leq 15 $ and $0\leq r <22 $. We suppose that $a_{5}>15$ for the last supposition. The inequality $\eqref{eq2.2}$, on the other hand, is where we have
\begin{gather*}
	\alpha^{r}>-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}\\>-11^{0.3a_{5}}-11^{0.5a_{5}}-11^{0.7a_{5}}-11^{0.85a_{5}}+11^{a_{5}}\\
	>11^{a_{5}}\left(-11^{-0.7a_{5}}-11^{-0.5a_{5}}-11^{-0.3a_{5}}-11^{-0.15a_{5}}+1\right)\\>0.9\cdot11^{a_{5}},
\end{gather*}
This indicates that $a_{5}<0.74r+0.04$ and $r>1.36a_{5}-0.06$ yields  $a_{5}<r$.\\

We utilise Theorem $\ref{th2.2}$, on the left hand side inequality $\eqref{eq3.1}$, with parameters $t:=3$, we take $\gamma_{1}:=\alpha, \gamma_{2}:=11, \gamma_{3}:=4\sqrt{2}$ and $b_{1}:=r, b_{2}:=-a_{5}, b_{3}:=-1$. We know that, $\gamma_{1},\gamma_{2},\gamma_{3} \in\mathbb{Q}(\sqrt{2})$. Thus $D:=[\mathbb{Q}(\sqrt{2}), \mathbb{Q}]=2$. To show that the left hand side inequality $\eqref{eq3.1}$, in nonzero, assume that $\Lambda_{1}=0$, which is implies that $32\cdot11^{2a_{5}}=\alpha^{2r}$, so $32\cdot11^{2a_{5}}\in \mathbb{Z}$ but $\alpha^{2r}\notin \mathbb{Z} $ for all value $r$ which is false. Note that 
$h\left(\gamma_{1}\right)=\frac{\log (\alpha)}{2},	h\left(\gamma_{2}\right)=\log (11)$ and $h\left(\gamma_{3}\right)=\log(4)+\frac{\log(2)}{2}$. Thus we can take
$A_{1} :=1.77, A_{2} := 4.8$ and $A_{3} := 3.5$. As we only get $\max\{|r|,|-a_{5}|,|-1|\}=r$, then we get  $B:=r$.
So by Theorem $\ref{th2.2}$, we get
\begin{equation}\label{eq3.2}
	\frac{5}{11^{0.15a_{5}}}>\left|\Lambda_{1}\right|>\exp \{-G(1+\log (r))(1.77)(4.8)(3.5)\},
\end{equation}
where $G=(1.4)\left(30^{6}\right)\left(3^{4.5}\right)(4)(1+\log (2))$. Since we have  $a_{5}>0.74r-0.74$, taking logarithm for both sides $\eqref{eq3.2}$, we get 
\begin{equation}
	0.27r-1.9<2.883622871\cdot10^{13}(1+\log(r)).\nonumber
\end{equation}
Thus
\begin{equation}
	r <3.9\cdot10^{15}.
\end{equation}
\subsection{Reducing the bound on r }
The upper bound for $r$ is given in the previous subsection $\ref{sub 1}$. Applying Lemma $\ref{lem2.3}$ allows us to lower this upper bound.\\
Let
\begin{equation}
	w_{1}:=r \log (\alpha)-a_{5} \log (11)-\log(4\sqrt{2}).\nonumber
\end{equation}
Then we have, by inequality $\eqref{eq3.1}$,
\begin{equation}
	\left|e^{w_{1}}-1\right|<\frac{5}{11^{0.15a_{5}}}.\nonumber
\end{equation}
We know $w_{1} \neq 0$, since $\Lambda_{1} \neq 0$.\\
If $w_{1}>0$, then
\begin{equation}
	0<w_{1}<\frac{5}{11^{0.15a_{5}}}.\nonumber
\end{equation}
And if $w_{1}<0$, since we have $\frac{5}{11^{0.15a_{5}}} < \frac{1}{2}$, for all $a_{5} > 15$, we get $\left|e^{w_{1}}-1\right|<\frac{1}{2}$. Therefore we get $e^{\left|w_{1}\right|}<2$, since we have $w_{1}<0$, thus:
\begin{equation}
	0<\left|w_{1}\right|<e^{\left|w_{1}\right|}-1\leq e^{\left|w_{1}\right|}\left|e^{\left|w_{1}\right|}-1\right|<\frac{10}{11^{0.15a_{5}}},\nonumber
\end{equation}
as a result, we have
\begin{equation}
	0<\left|r\log (\alpha)-a_{5} \log (11))-\log(4\sqrt{2})\right|<\frac{10}{11^{0.15a_{5}}}.\nonumber
\end{equation}
Both sides dividing by $\log (11)$, we get
\begin{equation}\label{eq3.4}
	0<\left|\frac{r\log (\alpha)}{\log(11)}-a_{5}-\frac{\log(4\sqrt{2})}{\log(11)}\right|<\frac{4.2}{11^{0.15a_{5}}}.
\end{equation}
As stated by the Theorem $\ref{th2.2}$, we have
\begin{gather*}
	\gamma:=\frac{\log (\alpha)}{\log (11)}, \mu:=-\frac{\log\left(4\sqrt{2}\right)}{\log(11)},\\ K:=4.2,L:=1.4, w:=a_{5}.
\end{gather*}
We recognise that  $\gamma$ is irrational. Let $\frac{p_{k}}{q_{k}}$ be the $k$ th convergence of $\gamma$. Also, we can take that $M=3.9\cdot10^{15}.$ We obtain $q_{31}=39964073819961229$, the denominator 31th convergent of using Maple programming, which is greater than 6M, and
\begin{equation}
	\varepsilon:=\|\mu q\|-M\|\gamma q\|=0.31.\nonumber
\end{equation}
Therefore, we get $a_{5}<122$, and hence 
\begin{equation}
	r<1.37*(122)+1<169,\nonumber
\end{equation}
Therefore, the equation $(r,a_{1},a_{2},a_{3},a_{4},a_{5})$ has a solution in the  $a_{5}$ and $r$ range shown above. Once more, we may use Theorem $\ref{th2.2}$, take  $M:=169$, and perform the calculation using Maple programming to arrive at $q_{11}$, the denominator's 11th convergent of  $\gamma$ , which is more than $6M$, we obtain
\begin{equation}
	\varepsilon:=\|\mu q\|-M\|\gamma q\|=0.06.\nonumber
\end{equation}
We get that $a_{5}<40$, and hence  
\begin{equation}
	r<1.37\cdot40+1<56.\nonumber
\end{equation}
In the range, we employ Maple programming $40\geq a_{5}\geq \max\{a_{1},a_{2},a_{3},a_{4}\}\geq0 $ and $r<56$. We obtain the same set of solutions in Theorem $\ref{th2.4}$. by programme inspection.
The proof is so finished.

\section{Proof theorem $\ref{th2.5}$}
\subsection{Bounding s}\label{sub 2}
When the equations $\eqref{eq2.3}$ and $\eqref{eq2.10}$ are combined, we obtain
\begin{equation}
	\frac{\lambda^s}{2}-11^{a_{5}}=-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}-\frac{\delta^{s}}{2},\nonumber
\end{equation}
Taking absolute values for both sides and $|\delta|^{s}<\frac{1}{2}$, we get
\begin{equation}
	\left|\frac{\lambda^s}{2}-11^{a_{5}}\right|<2^{a_{1}}+3^{a_{2}}+5^{a_{3}}+7^{a_{4}}+\frac{1}{4},\nonumber
\end{equation}
Dividing both sides by $11^{a_{5}}$ thus
\begin{equation}
	\left|\frac{\lambda^{s}11^{-a_{5}}}{2}-1\right|<\frac{2^{a_{1}}}{11^{a_{5}}}+\frac{3^{a_{2}}}{11^{a_{5}}}+\frac{5^{a_{3}}}{11^{a_{5}}}+\frac{7^{a_{4}}}{11^{a_{5}}}+\frac{1}{4*11^{a_{5}}},\nonumber
\end{equation}
yields
\begin{equation}\label{eq4.1}
	\left|\frac{\lambda^{s}11^{-a_{5}}}{2}-1\right|<\frac{5}{11^{0.15a_{5}}}.
\end{equation}
When $s$ and $a_{5}$ are related, we now want to estimate that relationship. Using the equation $\eqref{eq2.10}$ and the inequality  $\eqref{eq2.4}$, we are able to arrive at
\begin{equation}
	\lambda^{s-1} < q_{s}<11^{a_{5}},\nonumber
\end{equation}
which yields  $a_{5}>0.37s-0.37$ and $s<2.7a_{5}+1$. If $a_{5} \leq 10 $, then $s< 28 $. In order to find the set solutions in Theorem $\ref{th2.5}$, we use Maple programming to search in $0\leq a_{5}\leq 10 $ and $0\leq s <28 $. We presume that $a_{5}>10$ as our final presumption. With respect to the second inequality of $\eqref{eq2.4}$, we have
\begin{gather*}
	\lambda^{s}>-2^{a_{1}}-3^{a_{2}}-5^{a_{3}}-7^{a_{4}}+11^{a_{5}}\\>-11^{0.3a_{5}}-11^{0.5a_{5}}-11^{0.7a_{5}}-11^{0.85a_{5}}+11^{a_{5}}\\
	>11^{a_{5}}\left(-11^{-0.7a_{5}}-11^{-0.5a_{5}}-11^{-0.3a_{5}}-11^{-0.15a_{5}}+1\right)\\>0.97\cdot11^{a_{5}},
\end{gather*}
which implies that $a_{5}<0.37r+0.012$ and $s>2.7a_{5}-0.03$, and this yields  $a_{5}<s$.\\
We apply Theorem $\ref{th2.2}$, on the left hand side inequality $\eqref{eq3.1}$, we have $t:=3$, we take $\gamma_{1}:=\lambda, \gamma_{2}:=11, \gamma_{3}:=2$ and $b_{1}:=s, b_{2}:=-a_{5}, b_{3}:=-1$. Hence, we get, $\gamma_{1},\gamma_{2},\gamma_{3}\in \mathbb{Q}(\sqrt{2})$. Thus $D:=[\mathbb{Q}(\sqrt{2}), \mathbb{Q}]=2$. According to Theorem $\ref{eq2.2}$, the left hand side inequality $\eqref{eq4.1}$ is nonzero, assuming that $\Lambda_{2}=0$,which implies that $4\cdot11^{2a_{5}}=\lambda^{2s}$, so $4\cdot11^{2a_{5}}\in \mathbb{Z}$ but $\lambda^{2s}\notin \mathbb{Z} $ for all value $s$ which is false. We have 
$h\left(\gamma_{1}\right)=\frac{\log (\lambda)}{2},	h\left(\gamma_{2}\right)=\log (11)$ and $h\left(\gamma_{3}\right)=\log(2)$. Thus, we get
$A_{1} :=0.88, A_{2} := 4.8$ and $A_{3} := 1.4$. Note that if $\max\{|s|,|-a_{5}|,|-1|\}=s$, then we can take $B:=s$.
So by Theorem $\ref{th2.2}$, we obtain
\begin{equation}\label{eq4.2}
	\frac{5}{11^{0.15a_{5}}}>\left|\Lambda_{2}\right|>\exp \{-G(1+\log (s))(0.88)(4.8)(1.4)\},
\end{equation}
where $G=(1.4)\left(30^{6}\right)\left(3^{4.5}\right)(4)(1+\log (2))$. Since we have  $a_{5}>0.37s-0.37$, taking the logarithm for both sides $\eqref{eq4.2}$, we get
\begin{equation}
	0.13s-1.74<5.734662432\cdot10^{12}(1+\log(s)).\nonumber
\end{equation}
Thus
\begin{equation}
	s <1.6\cdot10^{15}.
\end{equation}
\subsection{Reducing the bound on s}
The upper bound for $s$, which was obtained in the previous subsection $\ref{sub 2}$, is too large. Applying Lemma $\ref{lem2.3}$, we lower this upper bound.\\
Let
\begin{equation}
	y_{1}:=s \log (\lambda)-a_{5} \log (11)-\log(2).\nonumber
\end{equation}
Then we have, by inequality $\eqref{eq4.1}$,
\begin{equation}
	\left|e^{y_{1}}-1\right|<\frac{5}{11^{0.15a_{5}}}.\nonumber
\end{equation}
We get $y_{1} \neq 0$, since $\Lambda_{2} \neq 0$.\\
If $y_{1}>0$, then
\begin{equation}
	0<y_{1}<\frac{5}{11^{0.15a_{5}}}.\nonumber
\end{equation}
And if $y_{1}<0$, since we have $\frac{5}{11^{0.15a_{5}}} < \frac{1}{2}$, for all $a_{5} > 10$, we get $\left|e^{y_{1}}-1\right|<\frac{1}{2}$. Therefore we get $e^{\left|y_{1}\right|}<2$, since we have $y_{1}<0$, thus:
\begin{equation}
	0<\left|y_{1}\right|<e^{\left|y_{1}\right|}-1\leq e^{\left|y_{1}\right|}\left|e^{\left|y_{1}\right|}-1\right|<\frac{10}{11^{0.15a_{5}}}.\nonumber
\end{equation}
Thus, we obtain
\begin{equation}
	0<\left|s\log (\lambda)-a_{5} \log (11))-\log(2)\right|<\frac{10}{11^{0.15a_{5}}}.\nonumber
\end{equation}
Both sides divided by $\log (11)$, we get
\begin{equation}\label{eq4.4}
	0<\left|\frac{s\log (\lambda)}{\log(11)}-a_{5}-\frac{\log(2)}{\log(11)}\right|<\frac{4.2}{11^{0.15a_{5}}}.
\end{equation}
Theorem $\ref{th2.2}$ states that we have 
\begin{gather*}
	\gamma:=\frac{\log (\lambda)}{\log (11)}, \mu:=-\frac{\log\left(2\right)}{\log(11)},\\ K:=4.2,, L:=1.4, w:=a_{5}.\nonumber
\end{gather*}
Let $\frac{p_{k}}{q_{k}}$ be the $k$-th convergence of the irrational number $\gamma$. We take that $M=1.6\cdot10^{15}.$ With the help of Maple programming, we get $q_{33}=39964073819961229$, the denominator 33th convergent of $\gamma$, which exceeds $6 M$, and 
\begin{equation}
	\varepsilon:=\|\mu q\|-M\|\gamma q\|=0.13.\nonumber
\end{equation}
As a result , we get $a_{5}<124$, and hence
\begin{equation}
	s<2.7\cdot(124)+1<336,\nonumber
\end{equation}
It implies that there is a solution for $(s,a_{1},a_{2},a_{3},a_{4},a_{5})$ in the $a_{5}$ and $s$ range mentioned before. Once more, we can use Maple programming to determine the denominator 11th convergent of $\gamma$, which is more than $6M$, using $M:= 336$ and Theorem $\ref{th2.2}$, we obtain 
\begin{equation}
	\varepsilon:=\|\mu q\|-M\|\gamma q\|=0.4.\nonumber
\end{equation}
We get that $a_{5}<36$, and hence
\begin{equation}
	s<2.7\cdot36+1<99.\nonumber
\end{equation}
We use Maple programming in the range $36\geq a_{5}\geq \max\{a_{1},a_{2},a_{3},a_{4}\}\geq0 $, and $s<99$. We obtain the same results from the programme checks for Theorem $\ref{th2.5}$.
As a result, the major outcome is finished
\section{Conclusion}
In order to find and reduce the upper bound considered above, we used a lower bound of linear forms in  logarithms of algebraic number and used Lemma \ref{lem2.3}. In the Theorems $\ref{th2.4}$ and $\ref{th2.5}$, we found all the answers to the Diophantine equations $\ref{eq2.9}$ and $\ref{eq2.10}$. 
\section{Development in the Future}
We point out that we could use our method to show that the Diophantine equation only has a finite number of solutions $B_{r}=2^{a_{1}}+3^{a_{2}}+5^{a_{3}}+7^{a_{4}}+11^{a_{5}},q_{s}=2^{a_{1}}+3^{a_{2}}+5^{a_{3}}+7^{a_{4}}+11^{a_{5}}$, in non-negative integers $a_{1},a_{2},a_{3},a_{4},a_{5}$ with $a_{5}\geq \max\{a_{1},a_{2},a_{3},a_{4}\}$. Future researchers will be faced with this dilemma, which we leave open.



\bigskip

\end{document}
